Author's response to reviews

Title: Evaluation of Coronary Blood Flow during Cardiac Arrest with Circulation Maintained through Mechanical Chest Compressions in a Porcine Model

Authors:

Henrik Wagner (henrik.wagner@skane.se)
Bjarne Madsen Hardig (bjarne.madsenhardig@jolife.com)
Stig Steen (stig.steen@med.lu.se)
Trygve Sjoberg (trygve.sjoberg@med.lu.se)
Jan Harnek (jan.harnek@skane.se)
Goran K Olivecrona (goran.olivecrona@med.lu.se)

Version: 9 Date: 15 December 2011

Author's response to reviews:

Lund, December 15, 2011

Dear Sir/Madam

We have now made the requested revisions of author affiliation and order of references and checked for typographical errors.

Please contact if there is anything else we can do.

Sincerely,

Goran Olivecrona, MD., PhD
Dept of Cardiology
Skane University Hospital
Lund University
221 85 Lund
SWEDEN

goran.olivecrona@med.lu.se